Author's response to reviews

Title: Incorporating the concept of Emotional Intelligence (EI) into a Personal and Professional Development (PPD) programme.

Authors:

Eva M Doherty (edoherty@rcsi.ie)
Patricia A Cronin (patcronin@rcsi.ie)
Gozie Offiah (gozieoffiah@rcsi.ie)

Version: 2 Date: 20 November 2012

Author's response to reviews: see over
20th November 2012

Dear Sir/Madam,

I am submitting this article titled “Incorporating the concept of Emotional Intelligence (EI) into a Personal and Professional Development (PPD) programme” to the journal; BMC Medical Education on behalf of all the authors involved namely, Eva Doherty, Patricia Cronin and Gozie Offiah.

This paper has not been submitted to any other journal and is not under consideration by any other journal. Also, this paper has not been published in any other journal. All authors have read the paper and agree to its content.

Kind regards,

Gozie Offiah.